Dear editors and reviewers:

Thank you very much for considering our revised manuscript entitled “Rituximab combined with BTKI to treat diffuse large B-cell lymphoma in elderly patients: Two case reports” (Manuscript NO: 86584 Case Report).

We appreciate the careful review and constructive comments provided by the reviewers of our manuscript. We have studied the comments carefully and provided answers in a point-by-point manner to each of the reviewers’ questions. We had made corrections which we hope are suitable for publication in World Journal of Clinical Cases. Below are our answers for questions raised by the reviewers.

**Comments 1:** Major 1. DeaAuthors, Thank you for sharing your experience. It would be better to check your manuscript with a native English colleague, or an application like Grammarly. You repeated half moon instead of two weeks or hlf a month!

Response: Thank you for your advice. We have revised the time.

**Editorial comments:**

**Comment 1:** Some suggestion: In “CASE PRESENTATION” section, it is suggested that chief complaints, history of present illness, history of past illness, personal and family history, physical examination, laboratory examinations, and imaging examinations should be described under a corresponding subheading according to the journal’s requirements. In the introduction, the incidence rate of epidemiology should be supplemented.

Response: Thank you for your advice. We have added chief complaints, history of present illness, history of past illness, personal and family history, physical examination, laboratory examinations, and imaging examinations according to the journal’s requirements. In the introduction, we have added the incidence rate of epidemiology should be supplemented.
Comment 2: Figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

Response: Thank you for your advice. We have revised the figure as required. We have arranged original figures and legends using PowerPoint detailed in the revision files.

Once again, thank you very much for your constructive comments and suggestions which would help us both in English and in depth to improve the quality of the paper.

Kind regards,

Min-Lei Zhao
Zhejiang Provincial Lishui Municipal Central Hospital
NO.289 Kuocang Road,
Hangzhou, Zhejiang, China
Zip code: 323000
E-mail: minleizhao@163.com